Author’s response to reviews

Title: Could caregiver reporting adherence help detect virological failure in early treated Human Immunodeficiency Virus-infected infants? Experience from the PEDIACAM study in Cameroon

Authors:

Francis Ateba Ndongo (atebfranc@gmail.com)
Josiane Warszawski (josiane.warszawski@inserm.fr)
Gaetan Texier (gaetex1@gmail.com)
Ida Penda (idapenda@yahoo.fr)
Suzie Tetang Ndiang (ndiangsuzie@yahoo.fr)
Jean-Audrey Ndongo (jeanndongo@yahoo.fr)
Georgette Guemkam (guemgeorg@hotmail.com)
Casimir Ledoux Sofeu (scl.ledoux@gmail.com)
Kfutwah Afumbom (kfutwah@pasteur-yaounde.org)
Albert Faye (albert.faye@rdb.aphp.fr)
Philippe Msellati (philippe.msellati@ird.fr)
Mathurin Cyrille tejiokem (temacy@yahoo.fr)

Version: 2 Date: 13 September 2014

Author’s response to reviews: see over
This manuscript deals with a pediatric aspect that is considered by *BMC Pediatrics Journal*. It is an original work that evaluates the diagnostic value of caregiver adherence reporting for the detection of virological failure to antiretroviral treatment in HIV-infected infants living in resource-poor settings where access to viral load exam is very weak. Its publication would help health care providers making decision in routine practice when the adherence questionnaire is used in the absence of viral load exam.

In this revised version of the manuscript, I included ethical consideration in the Methods section. I also included line and page numbers.

By Francis ATEBA NDONGO.